Author's response to reviews

Title: What are the factors associated with depressive symptoms among orphans and vulnerable children in Cambodia?

Authors:

Ken IC Ong (ken_ic_ong@hotmail.com)
Siyan Yi (ysiyan@khana.org.kh)
Sovannary Tuot (tsovannary@khana.org.kh)
Pheak Chhoun (cpheak@khana.org.kh)
Akira Shibanuma (shibanuma@gmail.com)
Junko Yasuoka (jyasuoka@m.u-tokyo.ac.jp)
Masamine Jimba (m jimba@m.u-tokyo.ac.jp)

Version: 5 Date: 12 July 2015

Author's response to reviews:

Dear reviewers,

Thank you very much for accepting our paper.

Sincerely,

Masamine Jimba